

Citation: Gold S, Donnelly CA, Woodroffe R, Nouvellet P (2021) Modelling the influence of naturally acquired immunity from subclinical infection on outbreak dynamics and persistence of rabies in domestic dogs. PLoS Negl Trop Dis 15(7): e0009581. https://doi.org/10.1371/journal. pntd.0009581

Editor: Daniel G. Streicker, University of Glasgow, UNITED KINGDOM

Received: November 26, 2020

Accepted: June 21, 2021

Published: July 20, 2021

Copyright: © 2021 Gold et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data and code are within the manuscript and its <u>Supporting</u> Information files.

Funding: SG is supported by the Imperial College London President's Scholarship (https://www. imperial.ac.uk/study/pg/fees-and-funding/ scholarships/presidents-phd-scholarships/) and Natural Environment Research Council (NE/ R011613/1). RW and SG were also supported by funding from Research England (https://re.ukri.org/ RESEARCH ARTICLE

Modelling the influence of naturally acquired immunity from subclinical infection on outbreak dynamics and persistence of rabies in domestic dogs

Susannah Gold^{1,2}*, Christl A. Donnelly^{2,3}, Rosie Woodroffe¹, Pierre Nouvellet⁴

1 Institute of Zoology, Zoological Society of London, London, United Kingdom, 2 MRC Centre for Global Infectious Disease Analysis, Department of Infectious Disease Epidemiology, School of Public Health, Imperial College London, London, United Kingdom, 3 Department of Statistics, University of Oxford, Oxfordshire, United Kingdom, 4 School of Life Sciences, University of Sussex, Falmer, Brighton, United Kingdom

* s.gold17@imperial.ac.uk

Abstract

A number of mathematical models have been developed for canine rabies to explore dynamics and inform control strategies. A common assumption of these models is that naturally acquired immunity plays no role in rabies dynamics. However, empirical studies have detected rabies-specific antibodies in healthy, unvaccinated domestic dogs, potentially due to immunizing, non-lethal exposure. We developed a stochastic model for canine rabies, parameterised for Laikipia County, Kenya, to explore the implications of different scenarios for naturally acquired immunity to rabies in domestic dogs. Simulating these scenarios using a non-spatial model indicated that low levels of immunity can act to limit rabies incidence and prevent depletion of the domestic dog population, increasing the probability of disease persistence. However, incorporating spatial structure and human response to high rabies incidence allowed the virus to persist in the absence of immunity. While low levels of immunity therefore had limited influence under a more realistic approximation of rabies dynamics, high rates of exposure leading to immunizing non-lethal exposure were required to produce population-level seroprevalences comparable with those reported in empirical studies. False positives and/or spatial variation may contribute to high empirical seroprevalences. However, if high seroprevalences are related to high exposure rates, these findings support the need for high vaccination coverage to effectively control this disease.

Author summary

Rabies-specific antibodies in unvaccinated domestic dogs have been reported in several studies, in some cases in a substantial proportion of the population. These antibodies may be the result of non-lethal rabies exposure leading to naturally acquired immunity, which could influence rabies transmission and persistence. In this study, a model was developed to consider a range of scenarios for naturally acquired immunity in a domestic dog

). CAD acknowledges funding from the MRC Centre for Global Infectious Disease Analysis (MR/ R015600/1), jointly funded by the UK Medical Research Council (MRC) and the UK Foreign, Commonwealth & Development Office (FCDO), under the MRC/FCDO Concordat agreement and is also part of the EDCTP2 programme supported by the European Union. CAD is also funded by the National Institute for Health Research (NIHR) Health Protection Research Unit in Emerging and Zoonotic Infections, a partnership between Public Health England, University of Liverpool, Liverpool School of Tropical Medicine and the University of Oxford (NIHR200907). Furthermore, CAD acknowledges funding from the Vaccine Efficacy **Evaluation for Priority Emerging Diseases** (VEEPED) grant (NIHR: PR-OD-1017-20002) from the National Institute for Health Research. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

population. Model outputs for rabies incidence and seroprevalence, the proportion of the population with rabies antibodies, were compared to estimates from empirical studies. Our results indicate that naturally acquired immunity could contribute to rabies persistence by limiting disease incidence and preventing population extinction. However, taking into account spatial structure and human response to high rabies incidence showed that rabies could persist without naturally acquired immunity, although immunity may contribute to the low incidence observed for endemic rabies. Assuming higher rates of subclinical exposure led to seroprevalences comparable to the estimates from empirical studies. However, uncertainty remains over the interpretation of rabies antibodies in unvaccinated individuals, and false positives may contribute to the high seroprevalences reported.

Introduction

Rabies is a zoonotic disease, caused by a neurotropic virus in the lyssavirus family. Despite eradication having been achieved in some parts of the world, the disease still presents a significant public health burden, particularly in rural Africa and Asia [1]. All mammalian species are susceptible to rabies, however only a limited number, primarily bats and carnivores, are able to maintain the virus within their populations [2]. In Africa, domestic dogs are the primary host of rabies and cause the majority of human cases, therefore controlling the disease in this species is key to preventing human rabies deaths [3].

A number of models have been constructed for rabies dynamics in domestic dog populations [4,5]. One assumption commonly used in these models is that immunity only occurs through vaccination, and not as a result of non-lethal exposure (for example [4–7], but see [8]). While rabies is usually fatal following the appearance of symptoms, Hampson et al. (2009) estimated that 51% of bite exposures in domestic dogs did not lead to clinical infection [9]. Of these exposure incidents, it is unclear whether the virus always fails to establish, and the host remains susceptible, or whether in some case the virus is cleared by the host's immune system, with subsequent development of protective immunity. While recovery from clinical rabies is rare, experimental studies have shown that non-lethal rabies exposure can occur, with exposed individuals showing no, or only minor, symptoms [10,11]. Under field conditions, there has been little consideration of seroconversion, the development of a specific antibody response, following rabies exposure. However, Cleaveland and Dye (1995) reported that of 17 dogs bitten by two suspected rabid individuals, 12 survived exposure of which four then seroconverted. As well as subclinical bite exposure, there is also limited evidence that oral exposure, for example from feeding on infected carcasses, can lead to development of rabies-specific antibodies in carnivores [12,13]. The potential for oral exposure to lead to development of rabies immunity is also supported by the success of vaccination campaigns using oral vaccines [14,15].

Rabies-specific antibodies have been detected in healthy, unvaccinated individuals across a number of domestic dog populations in rabies endemic areas, with a wide range of seroprevalences (the percentage of the population with detectable rabies-specific antibodies) reported [16]. There are a number of challenges to interpreting serology, including that different studies have used different tests and cut-offs to define seropositives [16,17]. However, given that clinical rabies typically occurs at low prevalence, affecting approximately 1% of dogs annually within populations where rabies is endemic [6,18–20], the high seroprevalences detected in some studies (e.g. 7.4% [21]; 28.0% [22]; 28.8% [23]; 30% [24]) could suggest high rates of non-lethal exposure relative to the rate of exposure leading to clinical infection. While in some

cases animals with no history of rabies exposure may test positive due to non-specific neutralisation or cross-reactivity, these high estimates of seroprevalence raise the possibility that naturally acquired immunity could play a more significant role in rabies dynamics than previously considered [16].

As a result of the low incidence of rabies in domestic dog populations, rabies seldom leads to substantial depletion of the population [19,25,26]. However, there is evidence that rabies transmission is frequency dependent, with transmission rates remaining relatively constant across a wide range of domestic dog densities [18,27]. Modelling suggests that this form of transmission should lead to high-prevalence outbreaks and substantial population losses [18]. Mechanisms that could limit rabies incidence under real-life conditions include spatial structure (which could lead to local depletion of the susceptible pool without widespread transmission), and human intervention, such as killing and isolation of infectious dogs, following increased incidence [5,9,18,28]. Naturally acquired immunity could potentially also contribute to the low-level persistence of rabies by protecting a proportion of the population, which can then produce new susceptible hosts. Naturally acquired immunity has been considered as a mechanism for the persistence of rabies in vampire bats [29], but is usually not considered in domestic dog models [e.g. 5,7,30].

In this study, we explore the implications of naturally acquired immunity to the rabies virus, resulting from subclinical exposure, for the dynamics of rabies in domestic dogs using a stochastic model parameterised for Laikipia County, Kenya. Rabies is endemic in Laikipia and a seroprevalence of 28% was previously reported in the domestic dog population [22]. A non-spatial model is initially used to explore a wide range of parameter values for naturally acquired immunity in domestic dogs. This model is then extended into a spatial model to consider a subset of potential immunity scenarios. While the spatial model relies on a greater number of assumptions, it allows for consideration of the implications of immunity under a more realistic approximation of rabies dynamics.

Methods

A stochastic model of rabies was developed, parameterised for the domestic dog population in Laikipia County, Kenya. A compartmental structure was used with dogs divided into susceptible (S) individuals which are able to contract the virus, exposed (E) individuals which are incubating the virus, infectious (I) individuals which are able to transmit the virus, and individuals with naturally acquired immunity (R) which are immune to re-infection and are assumed to have detectable rabies antibodies. A total population size of 63,434 dogs was simulated and it was assumed that all individuals in the population were unvaccinated. Methods for estimating domestic dog numbers and simulating dog demography are presented in the S1 Text.

Transmission dynamics

For rabies, the majority of transmission is through bite exposure [31]. However, other routes of transmission, such as through oral exposure, may be relevant when considering subclinical infection [13,32,33]. Exposure was therefore defined as any interaction between individuals that could result in viral transfer. The exposure rate was assumed to be frequency dependent, based on several studies which have indicated that R_0 (the basic reproduction number: the average number of secondary cases produced by one case in a completely susceptible population) for rabies is relatively consistent for domestic dog populations across a range of population densities [27,34,35].

Initial model exploration was conducted using an R_0 value of 1.2, which is in the range of one to two typically reported for domestic dog rabies [9,35]. The influence of higher R_0 values

within this range was also considered using the spatial model. The following equation for R_0 was rearranged to calculate the infectious exposure rate (β), the number of individuals exposed per day by an infectious individual:

$$R_0 = \beta \phi / v \tag{1}$$

The product of the exposure rate (β) and probability of developing clinical infection following exposure (ϕ) captures the rate of transmission. The basic reproduction number, R₀, is found by multiplying this quantity by the average duration of the infectious period (1/v).

Modelling naturally acquired immunity

Following exposure of a susceptible individual, we assumed there were three possibilities. Depending on the probability of developing clinical infection (ϕ), a proportion of individuals enter the exposed compartment, from which they progress to becoming clinically infectious at rate σ per day. Once infectious, the dogs succumb to rabies after 3.1 days (v^{-1}) on average [9]. The remaining proportion (1- ϕ) are subclinically exposed and either became immune through developing an antibody response (ρ), and enter the R compartment, or remain susceptible (1- ρ). This model structure makes the assumptions that dogs cannot develop symptomatic rabies and then recover and that individuals which are subclinically exposed do not transmit the disease.

In this model, it was assumed that any individuals in the immune (R) compartment had detectable rabies antibodies which confer protective immunity. Therefore, the proportion of individuals in the R compartment was considered to be equal to the predicted seroprevalence. We assumed the dynamics of antibodies developed through natural exposure would be comparable to those from vaccine-derived immunity. Based on the rate at which antibody titres become undetectable under field conditions following vaccination, for initial model exploration it was assumed individuals would remain in the R compartment for one year on average (δ^{-1}) [36–38]. However, challenge experiments have shown that rabies immunity can persist for longer durations, and dogs may remain protected even where antibody tires have waned below detectable levels [39–41], therefore the influence of a longer persistence time of three years was also explored using the spatial model.

The following equations describe the full non-spatial model:

$$\frac{dS}{dt} = a(S+R) - \frac{\beta SI}{N} \left((1-\phi)\lambda + \phi \right) + \delta R - \mu S$$
⁽²⁾

$$\frac{dE}{dt} = \frac{\beta SI}{N}\phi - \sigma E \tag{3}$$

$$\frac{dI}{dt} = \sigma E - \nu I \tag{4}$$

$$\frac{dR}{dt} = \frac{\beta SI}{N} (1 - \phi)\lambda - \delta R - \mu R \tag{5}$$

$$a = \begin{cases} a_max \frac{1-qN}{K}, & \text{if } N < K/q \\ 0, & \text{otherwise} \end{cases}$$
(6)

Parameters values are described in Table 1.

Epidemiological description	Symbol	Value	Notes	Source
Basic reproduction number	R ₀	1.2	Higher values of 1.5 and 2 were also explored using the spatial model.	
Maximum birth rate	a_max	0.00302 per dog per day	Data from Laikipia County domestic dog population for sex ratio, litter [size and litters per year.	
Background death rate	μ	0.00123 per dog per day	Assumes average lifespan of 2.2 years	
Scaling parameter for relationship between birth rate and population size	q	0.593	Parameterised so birth rate equal to death rate at carrying capacity.	
Exposure rate per infectious individual per day	β	-	Calculated from R_0, φ and $\mu.$	
Probability of developing clinical infection after exposure	φ	-	Varied for model exploration between 0.05 and 0.95.	
Probability of developing protective immunity following subclinical exposure	ρ	-	Varied for model exploration between 0 and 1.	
Rate of progression to clinical infection	σ	0.0448 per exposed dog per day	Reciprocal of average latent period (22.3 days). [9	
Rate of waning immunity	δ	0.00274 per immune dog per day	Assumed average duration of naturally acquired immunity of 1 year. A longer duration of 3 years is also explored using the spatial model.	
Rate of rabies mortality	ν	0.323 per infectious dog per day	Reciprocal of average infectious period (3.1 days). [9]	
Total carrying capacity and starting population	К	63,434 dogs	Estimated from human-to-dog ratio.	

Table 1. Non-spatial model parameter values. Sensitivity analysis was conducted for all parameters in the non-spatial model, excluding R_0 and K which were kept constant, and q and β , which were calculated from other parameters. Sensitivity analysis results are presented in the S2 Text.

https://doi.org/10.1371/journal.pntd.0009581.t001

Incorporating spatial structure and human intervention

While the non-spatial model allows exploration of rabies dynamics while minimising the number of assumptions made, it fails to capture key aspects of rabies dynamics. Previous studies developing rabies epidemiological models have shown the importance for rabies dynamics of spatial structure, human-mediated movement of dogs and human intervention in the case of high rabies incidence [18,43–45]. To consider the implications of naturally acquired immunity under a more realistic scenario, the non-spatial model was extended to incorporate spatial heterogeneity and human intervention.

A patch structure was used, a form of metapopulation model in which sub-populations represent adjacent parcels of land. In total 154 patches were included, representing a range of land uses, such as individual ranches, villages and communities which have distinct human communities and are therefore expected to have discrete domestic dog populations (Fig 1).

Infectious dogs were assumed to remain within their patch but could transmit to susceptible individuals in other patches. The force of infection within patch *i* was therefore assumed to depend on the number of infectious individuals, multiplied by the probability of contact with the patches those infectious individuals were in. Details for calculating contact probabilities are provided in the S1 Text. While infectious dogs were assumed to remain in their patch, movement of susceptible (S), exposed (E) and immune (R) dogs between patches by human-mediated movement was simulated using a gravity model (See S1 Text for details). Human intervention in response to high rabies incidence through interventions such as tying up or killing rabid dogs has been suggested to be an important factor in limiting rabies incidence [9,18]. This response was incorporated into the model through an incidence-dependent increase in the mortality rate of infectious dogs, assumed to act at a local level within patches. Within each patch, it was assumed that if more than 1% of the carrying capacity of the patch died from rabies in the previous month, increased human intervention caused an increase in



Fig 1. Map of Laikipia County showing underlying human density and patches used in the spatial model. Human population per square kilometre for Laikipia county and the surrounding areas is shown, based on population counts available at https://sedac.ciesin.columbia.edu/data/set/gpw-v4-population-count-rev11 [46]. Grey lines indicate patch boundaries with the thicker line indicating the border of Laikipia County. Patch boundaries were modified from a shapefile of property boundaries from Mpala Research Centre, Laikipia County, Kenya [47].

https://doi.org/10.1371/journal.pntd.0009581.g001

the death rate of infectious individuals. Further detail on parameterisation of this response is provided in the <u>S1 Text</u>, and results are presented from simulations across a range of levels of intervention to show the influence on model outputs in <u>S1 Fig</u>.

Immunity scenarios

The parameters of key interest for the model are the proportion of exposures leading to clinical infection (ϕ) and, of subclinical exposures (1- ϕ), the proportion which develop immunity (ρ). Of bite exposures, Hampson et al. (2009) estimated that 0.49 led to clinical infection. However, other routes of exposure, such as feeding on infected carcasses, or saliva transfer during social contact, may be less likely to lead to clinical infection relative to bite exposure [31,33]. Compared to considering only bite exposures, incorporating other forms of exposure could therefore lead to a higher exposure rate (β) but lower probability of developing clinical infection (ϕ).

Using the non-spatial model, to explore the implications of different probabilities of nonlethal rabies exposure, we varied ϕ at five levels between 0.05 (very few exposures lead to clinical infection) and 0.95 (almost all exposures lead to clinical infection) but fixed R₀ at 1.2. As a result, when ϕ was higher, the exposure rate (β) was lowered according to the relationship R₀ = $\beta \phi/v$, to produce the same number of secondary clinical infections. This relationship gave a range for β from 0.37–7.06 exposures per infectious individual per day for an R₀ of 1.2. For each level of ϕ , we ran the model across five levels of ρ , the probability of developing immunity, from 0 (no subclinical exposures lead to immunity) to 1 (all subclinical exposures lead to immunity) to give 25 parameter combinations in total. In addition, a sensitivity analysis of the non-spatial model is presented in the S2 Text. Following exploration of the full range of parameter values, three parameter combinations were considered in detail, representing three potential scenarios (Table 1). The first scenario (A) assumed no development of naturally acquired immunity. In this scenario, it was assumed transmission only occurred through bite exposures and 50% of bites led to clinical infection (Based on estimate of 0.49 from Hampson et al. (2009), $\phi = 0.5$) with those individuals not developing clinical infection remaining susceptible ($\rho = 0$). The second scenario (B) assumed immunity could be acquired following non-lethal bite exposure. The probability of clinical infection was equal to the first scenario ($\phi = 0.5$) but of bite exposures which did not lead to clinical infection, it was assumed that 25% developed immunity ($\rho = 0.25$). The third scenario (C) represents the least conservative scenario for naturally acquired immunity. It was assumed that, in addition to bite exposure, other forms of exposure also occurred, such as oral exposure through social contact, which increased the exposure rate, but a lower proportion of exposures developed clinical infection ($\phi = 0.05$ and $\beta = 7.7$). Of subclinical exposures, we assumed 50% developed immunity ($\rho = 0.5$).

Running model and outputs extracted

The model was implemented using the SimInf package in R [48]. Infection dynamics were implemented as continuous-time Markov chains using the Gillespie stochastic algorithm. The model was run with a daily time step with outputs extracted at weekly intervals. For each simulation of the spatial model, the model was initiated with a prevalence of 0.5% (24 dogs) in the patch with the largest population size. In the non-spatial model, 24 individuals were also introduced. For each parameter combination, 1000 simulations were run for 30 years.

Probability of rabies persistence was measured as the percentage of simulations in which infectious individuals remained present in the population at 30 years post-introduction. For simulations in which rabies remained present, the annual incidence per 100,000 dogs for the final year, and the immune proportion in the final time step across the population were extracted. In addition, to consider spatial variation in predicted seroprevalence between patches in the spatial model, serology sampling within-patches was simulated. For each simulation, a patch was randomly selected and sampling of 30 individuals simulated. The proportion of these individuals in the R compartment was taken as the within-patch predicted seroprevalence.

Empirical estimates for rabies incidence, population decline and seroprevalence

To consider the plausibility of different parameter combinations for non-lethal exposure and naturally acquired immunity, we compared model outputs for rabies incidence, population decline, and predicted seroprevalence to empirical estimates extracted from the literature. For Laikipia County, there is currently no estimate of annual rabies incidence in the domestic dog population. Therefore, in the absence of location-specific data, we extracted plausible ranges from other free-ranging domestic dog populations (see Table 2). We extracted these data from studies conducting active surveillance, as passive surveillance is likely to significantly underestimate incidence [26]. Based on these studies, we assumed an upper limit for annual rabies incidence of 1,500/100,000 (Table 2).

Due to the low incidence of rabies, substantial population decline is not expected to result from endemic rabies, although populations may fluctuate in size. For example, in dog populations with endemic rabies in Indonesia and South Africa, Morters et al. (2014) report variation in population size of up to 22% from the mean over the study period. In South Africa, Conan et al. (2015) report annual changes in population size from +18.6% to -24.5%. In this study we

Scenario	Exposure rate per infectious individual (β)	Probability of developing clinical infection (φ)	Probability of developing immunity (ρ)	Probability that exposure leads to immunity $((1-\phi)\rho)$
A- No immunity	0.79 per day	0.5	0	0
B- Low immunity	0.79 per day	0.5	0.25	0.1275
C- High immunity	7.74 per day	0.05	0.5	0.1875

Table 2. Parameter values for naturally acquired immunity scenarios A, B and C.

For each of these three immunity scenarios, the influence of a longer persistence time for naturally acquired immunity ($\delta^{-1} = 3$ years) and higher R₀ values of 1.5 and 2 were also explored.

https://doi.org/10.1371/journal.pntd.0009581.t002

assumed that once endemic, taken as 30-years post-introduction, rabies would not cause population decline greater than 20% relative to the carrying capacity.

We also considered the seroprevalence predicted by the model relative to empirical estimates. In Laikipia, Prager et al. (2012) reported a seroprevalence of 28% in 75 domestic dogs tested using a rapid fluorescent focus inhibition test (RFFIT; 95% CI: 18.2–39.6%). However, this study used a low cut-off of 0.05 IU/mL which increases the probability of false positives. There is also evidence that enzyme-linked immunosorbent assays (ELISAs) are more specific for detecting non-lethal exposure relative to neutralisation tests [21]. Studies using ELISAs have also detected high sero-prevalences. Laurenson et al. (1997) found a seroprevalence of 30.0% in Namibia [24], Bahloul et al. (2005) of 28.8% in Tunisia [23] and Cleaveland et al. (1999) of 7.4% in Tanzania [21]. We therefore considered a range of 7–30% for seroprevalence in domestic dogs [16].

Results

Non-spatial model

Assuming an R₀ of 1.2, with no spatial structure and no incidence-dependent human response, rabies was not predicted to persist in the absence of any naturally acquired immunity ($\rho = 0$) or with a low probability of sub-clinical exposure ($\phi = 0.95$) (Fig 2A). Allowing a proportion of



Fig 2. Results from non-spatial model of effects of non-lethal exposure (ϕ) and acquired immunity (ρ) on probability of rabies persistence, median percentage population decline, median annual incidence per 100,000 and median predicted percentage seroprevalence. Plot a presents the median values for the full range of parameter combinations. Plot b shows a subset of parameter combinations (Scenarios A, B and C) indicated by labels on plot a with median values and interquartile ranges over the full 30 years for total population size, annual incidence per 100,000 dogs and seroprevalence (%). Grey bands indicate the range of values considered plausible based on empirical estimates, as shown in Table 3.

https://doi.org/10.1371/journal.pntd.0009581.g002

dogs to develop immunity ($\rho > 0$) increased the probability of rabies persistence, with the highest persistence probabilities at intermediate probabilities of acquired immunity. For example, assuming 50% of exposures led to clinical infection, with 25% of sub-clinically exposed individuals developing immunity (Scenario B- $\phi = 0.5$, $\rho = 0.25$), rabies remained endemic in 94% of the simulations. The median annual incidence under this scenario was high at 61,820/ 100,000 (Interquartile range (IQR): 19,139/100,000), with a high population decline relative to the carrying capacity (Median: 96.0%, IQR: 2.0%). Assuming higher probabilities of acquired immunity led to lower persistence, due to reduced incidence and an increased probability of stochastic extinction. For example, under scenario C, with a low probability of an individual developing clinical infection ($\phi = 0.05$), and a high probability of developing immunity ($\rho =$ 0.5), incidence was reduced to 2,207/100,000 (IQR: 1181) with rabies persisting in 87% of simulations. No parameter combination in the homogenous model led to median outputs for both incidence and population decline in the range considered plausible (Table 3). However, of the parameter combinations in which rabies was predicted to persist in the majority of simulations, Scenario C (the high immunity scenario, Table 2) had incidence closest to the plausible range and a low median population decline of 3.6% (IQR: 1.1%) (Fig 2).

Of the full set of 25 parameter combinations considered, the combination which generated the highest median predicted seroprevalence, of 16.0% (IQR: 3.5%), was a probability of an individual developing clinical infection (ϕ) of 0.05 and a probability of developing immunity (ρ) of 0.25. This predicted seroprevalence is close to the threshold for herd immunity for an R₀ of 1.2 (1-1/R₀ = 0.167). At this threshold each infectious individual infects on average one other, leading to stable endemic infection. For the subset of three immunity scenarios considered, predicted seroprevalences using the non-spatial model were 0.0% (IQR: 0.0%), 11.1% (IQR: 5.2%) and 14.9% (IQR: 6.1%) for scenarios A, B and C respectively.

Incorporation of spatial structure and incidence-dependent human response

The non-spatial model was extended to a spatial patch model including human-mediated movement of dogs and incidence-dependent human response to explore the three immunity scenarios (Fig 3). Results from spatial model formulations excluding human-mediated movement and incidence-dependent human response are presented in S2 and S3 Figs.

Incorporating these additional assumptions resulted in an increase relative to the non-spatial model in the probability of persistence of rabies in the absence of immunity, with the disease persisting in 82.9% of simulations for scenario A. The persistence probability associated with scenario B, with low levels of naturally acquired immunity was marginally lower, at 81.9%. In the high immunity scenario, C, rabies did not persist in any simulation at this R₀ value. Scenario C assumed a high transmission rate and for every exposure leading to clinical

Outputs	Assumed plausible range	Literature estimates and sources	
Population decline relative to carrying capacity	0-20%	[49,50]	
Annual incidence per 100,000 dogs	<1500/100,000 dogs, allowing	860/100,000 - [26]	
	for under-reporting	1089/100,000 - [20]. For unvaccinated villages where active surveillance conducted.	
		412/100,000 - [19]	
Seroprevalence	7-30%	[16]	

Table 3. Assumed plausible ranges of population decline, annual incidence per 100,000 dogs and seroprevalence for comparison to model outputs.

https://doi.org/10.1371/journal.pntd.0009581.t003



Spatial model, R0=1.2, Duration of immunity= 1 year



https://doi.org/10.1371/journal.pntd.0009581.g003

infection ($\phi = 0.05$), 9.5x more become immune ((1- ϕ) ρ where $\rho = 0.5$). As a result, rapid depletion of the susceptible population within patches reduced the probability of persistence.

For both scenarios A and B, median incidence was higher than the range considered plausible, however incidence was lower in the low immunity scenario (B) relative to the no immunity scenario (A), with a median of 5,566/100,000 (Scenario A, IQR: 2,583/100,000) relative to 7,810/100,000 (Scenario B, IQR: 3,485/100,000; Fig 3). Median population decline for scenario B was within the plausible range, at 15% (IQR: 3%), whereas without immunity (Scenario A) decline was greater at 22% (IQR: 5%). In the low immunity scenario (B), seroprevalence at a population-wide level was predicted to be low, with a median of 0.97% (IQR:0.42%). Incorporating low levels of naturally acquired immunity therefore lowered the predicted incidence and mitigated the population decline, despite a low population-wide predicted seroprevalence.

The duration of naturally acquired immunity for rabies has not been established, and primary analyses assumed an average duration of 1 year, based on the persistence of detectable antibodies following vaccination from field studies [36–38]. However, increasing this duration to three years ($\delta = 1/1095$), did not substantially change the model outputs (Fig 4). For scenario B, increasing the duration of immunity increased median seroprevalence only slightly, from 0.97% (IQR:0.42%) to 1.5% (IQR:0.7%). Due to the rapid turnover in domestic dog populations, with the average lifespan assumed to be 2.2 years (μ^{-1}), seropositive dogs are likely to die before immunity wanes, therefore limiting the influence of longer persistence of naturally acquired immunity.

Assuming higher R_0 values of 1.5 and 2 led to substantial changes in the model outputs (Fig 4). Increasing R_0 led to a higher persistence probability for the high immunity scenario, C, with rabies persisting in 13.8% of simulations for an R_0 of 1.5 and 89.3% for an R_0 of 2, relative to 0.0% for an R_0 of 1.2. In simulations for scenario C where rabies did persist, the median



Fig 4. Comparison of spatial model outputs with increased average duration of naturally acquired immunity ($\delta^{-1} = 3$ years) and higher R₀ values (R₀ = 1.5 and R₀ = 2), compared to the default model with R₀ = 1.2 and a 1-year average duration of immunity. Boxplots are shown for the outbreak time (a) (time from introduction to no infectious individuals remaining, or end of simulation), population decline relative to the carrying capacity (b), incidence per 100,000 dogs in year 30 post introduction (c) and seroprevalence at the end of the simulation (d). Grey bands indicate the range of values considered plausible based on empirical estimates, as shown in Table 3.

https://doi.org/10.1371/journal.pntd.0009581.g004

incidence for an R₀ of 1.5 was 1141/100,000 (IQR: 631/100,000) and population decline was 6.5% (IQR: 0.8%), consistent with the ranges considered plausible (Fig 4). For an R₀ of 2, the incidence was higher at 3904/100,000 (IQR: 1313/100,000), however the population decline remained within the plausible range at 8.9% (IQR: 1.7%). The median predicted seroprevalence for these scenario C simulations was 6.8% (IQR: 3.8%) for an R₀ of 1.5, and 26.1% (IQR: 8.1%) for an R₀ of 2. For scenarios A and B, increasing R₀ to 1.5 led to higher incidence and greater population decline, further increasing these outputs above the levels considered

plausible (Fig 4). Further increasing R_0 to 2 led to rabies not persisting under the no immunity scenario (A) in any simulations, and only persisting in 20.5% of simulations under the low immunity scenario (B).

Spatial variation in predicted seroprevalence

At a population level, predicted seroprevalences for an R_0 of 1.2 were low. However, substantial spatial variation in seroprevalence between patches occurred, as shown in Fig 5. Simulating sampling of 30 individuals from a single randomly selected patch for each of the simulations for scenario B gave a median predicted sample seroprevalence of 0. However, for the 11.1% of simulations in which seropositives were detected in the sample, the median predicted seroprevalence was 3.3%, with a maximum of 13.3%, relative to the population level seroprevalence of 0.97% (IQR:0.42%). Therefore, depending on the distribution of sampling, sample seroprevalences may vary substantially from the population-level seroprevalence.



Fig 5. Spatial variation in seroprevalence between patches from model simulations for scenario B, with $R_0 = 1.2$ and $\delta = 1/365$. a) Maps of Laikipia County showing seroprevalence in each patch at 30 years post introduction from three randomly selected simulations. Patch boundaries were generated from a shapefile of property boundaries from Mpala Research Centre, Laikipia County, Kenya [47]. b) Boxplots showing variation in seroprevalence in all patches at 30 years post introduction. The grey band indicates the range of seroprevalence considered plausible based on empirical estimates, as shown in Table 3. The red dotted line indicates the median population-level seroprevalence.

https://doi.org/10.1371/journal.pntd.0009581.g005

Discussion

Immunity in models of canine rabies is typically only considered in relation to vaccination. However, rabies-specific antibodies in unvaccinated individuals have been detected in a number of studies of domestic dog populations. While there is still debate over the importance of these antibodies [16], the high seroprevalences detected in some studies warrant consideration of their potential implications. In this study, a model of domestic dog rabies was developed to explore the influence of subclinical infection and immunity for rabies dynamics.

Exploration of the non-spatial model showed that, in the absence of other factors limiting incidence of rabies, naturally acquired immunity could play a role in stabilising rabies outbreaks. Without spatial structure and assuming rabies transmission was frequency dependent, with an R_0 value of 1.2, introduction of rabies was predicted to lead to depletion of the domestic dog population in the absence of immunity. In previous non-spatial models of rabies, population depletion has been prevented by assumptions of density-dependent transmission, or low R_0 values with high population growth [6,18,30]. Studies of the transmission dynamics of rabies have suggested these assumptions are potentially unrealistic, with evidence for higher R_0 values and frequency-dependent transmission [9,18,27,34,35]. Our model suggests that naturally acquired immunity can facilitate outbreak persistence under these assumptions by reducing incidence and preventing depletion of the susceptible population, as immune hosts produce susceptible offspring which can then become infected. However, having high rates of immunizing subclinical exposure could also push incidence to low levels, leading to an increased probability of stochastic fade out of the outbreak. The non-spatial model therefore predicted that endemic infection was most likely at intermediate levels of acquired immunity.

While the non-spatial model allowed exploration of the implications of acquired immunity with limited assumptions, it fails to capture key aspects of rabies dynamics. Previous studies have indicated the importance of spatial structure, human-mediated dog movement and human response to increased rabies incidence as important factors in rabies dynamics [4,18,28,51,52]. Results from the spatial model incorporating these factors showed that persistence of rabies could occur in the absence of immunity, as spatial structure and human intervention act to limit incidence and prevent population extinction. However, model predictions for annual incidence and population decline remained higher in the absence of immunity than empirical estimates. Including finer-scaled spatial structure and higher levels of human intervention could produce realistic estimates in the absence of immunity. For example, Beyer (2010) considered spatial structure on the scale of a single village of 288 dogs [28], considerably lower relative to the total carrying capacity of 63,434 dogs in this model. This fine-scale structure led to realistic outputs without inclusion of immunity. Furthermore, assuming a stronger response to high rabies incidence through human intervention also led to lower incidence in the absence of immunity (S2 Fig). Estimating the strength of this response is highly challenging. Hampson et al. (2009) found that in Tanzania, killing of infectious dogs reduced the infectious period by around 16%. However, how this response scales with incidence and varies temporally and spatially is unclear. Including this relationship prevents unrealistically high incidences, however as the strength of this effect has not been reliably estimated, its influence relative to other factors, such as spatial structure and immunity, is difficult to establish. While including finer scaled spatial structure or stronger human intervention could reduce incidence to levels within the range considered plausible, these factors do not account for the seroprevalences reported in empirical studies [16]. Including immunity for an R_0 of 1.2 led to lower incidences, despite low predicted seroprevalence relative to observed levels. Potentially, naturally acquired immunity, in combination with spatial structure and human intervention, may therefore act to dampen rabies outbreaks, leading to low-level endemic infection.

For an R_0 value of 1.2, none of the scenarios considered produced the levels of incidence, population decline and seroprevalence considered realistic. In the high immunity scenarios, rabies persistence was not predicted to occur due to depletion of the susceptible population. For an R₀ of 1.2, herd immunity within a patch is effectively reached once more than 16.7% (1- $1/R_0 = 0.167$). of the population is immune. Local susceptible depletion due to immunizing exposure can therefore lead to substantial reductions in transmission, despite low populationlevel seroprevalences. However, assuming higher R₀ values of 1.5 and 2 led to persistence of rabies in 13.8% and 89.3% of simulations respectively under the high immunity scenario, as the threshold to reach herd immunity is higher. For an R_0 of 1.5, the estimates of incidence and population decline under this scenario were also consistent with empirical estimates. Assuming higher transmission rates also led to higher median predicted population-level seroprevalences at 6.8% and 24.7% for R₀ values of 1.5 and 2 respectively. These estimates are closer to levels observed in empirical studies, for example Cleaveland et al. (1999) reported a seroprevalence of 7.4% in Tanzania and Bahloul et al. (2005) of 28.8% in Tunisia. The higher R_0 values remain within the range of one to two typically reported for canine rabies [9]. However, under the high immunity scenario, much higher rates of exposure, and lower probabilities of developing clinical infection, were assumed than are typically considered for canine rabies. For example, for an R_0 of 2 under the high immunity scenario, a rate of 12.9 exposures per infectious dog per day was assumed, with only 5% of these exposures leading to clinical infection. This parameterisation for the high immunity scenario is comparable to what has been assumed for lyssavirus dynamics in bats, in which the probability of developing immunity is considered to be much higher [29,53]. However, Hampson et al. (2009) estimated that on average, a rabid dog bites 2.15 others during its infectious period, of which 49% develop clinical infection, suggesting an exposure rate of 12.9 per day is unrealistically high if bites are considered the sole source of exposure. If the high empirical seroprevalences detected in some populations are due to higher rates of exposure, it may suggest that routes of transmission other than bite exposure are leading to development of an antibody response, such as through oral exposure during social contact or feeding on infected carcasses [12,13,54].

While high seroprevalences detected in empirical studies could reflect higher transmission rates, other factors may also be responsible. Substantial variation between serology tests has been shown, with evidence that neutralisation tests such as the rapid fluorescent focus inhibition test (RFFIT) may be less specific for detecting non-lethal exposures relative to ELISAs [21]. For example, the high seroprevalence of 28% detected in Laikipia, using the RFFIT and a low cut-off, may be partially explained by false positives [22]. However, even studies using ELI-SAs with higher cut-offs have found high seroprevalences [23,24]. For example, Cleaveland et al. (1999) found a seroprevalence of 7.4% in a domestic dog population in Tanzania using an ELISA and no false positives were detected on a rabies-free island using this test, suggesting high specificity. A further possibility is that cross-reactivity is occurring with other circulating lyssaviruses, although in the absence of further serological and surveillance data, this cannot be confirmed [16]. A further factor which may contribute to the discrepancy between observed seroprevalences and predicted population-level seroprevalence is spatial variation. As shown in Fig 5, predicted seroprevalences were not consistent across the landscape, therefore the population-wide predicted seroprevalence is expected to differ from sample seroprevalences. For example, localised sampling in areas where there has been a recent outbreak could lead to higher detected empirical seroprevalences. However, under the low-immunity scenario, the highest simulated sample seroprevalence was 13.3%, remaining lower than reported in some empirical studies [16]. Furthermore, empirical studies have in most cases used larger sample sizes, and in a range of locations, which should lead to a closer approximation of the population-level seroprevalence [e.g. 21,22].

A further challenge to the interpretation of rabies serology is that it is currently unknown whether antibodies from natural exposure confer protection to re-exposure [16]. There is limited evidence that previously unvaccinated seropositive individuals show an anamnestic response to vaccination consistent with immunity [15,23,38,55]. However, even in vaccinated individuals, serology status is not definitive proof of an effective immune response, and individuals with a detectable titre may still succumb to the disease [56]. If antibodies from non-lethal exposure provide no, or only partial protection, this would influence the expected dynamics. For example, partial immunity could potentially allow higher seroprevalences to occur within populations without the same reduction in transmission, potentially increasing the probability of persistence at higher seroprevalence levels.

The results of this study should be considered in the context of rabies surveillance and control. In this study we assumed all dogs were unvaccinated, to allow consideration of the influence of naturally acquired immunity on persistence in the absence of control. In Laikipia, mass vaccination has only been conducted since 2015 and prior to this point by far the majority of dogs would have been unvaccinated [57]. However, low levels of vaccination coverage may act in a similar way to naturally acquired immunity, dampening incidence of rabies and potentially promoting persistence. Kitala et al. (2002) modelled coverage of 24% in Machakos district, Kenya and found that this low level of coverage increased the stability of transmission and led to endemic establishment. While not within the scope of this paper, future work to include both vaccination and non-lethal exposure could lead to further insights into the influence of naturally acquired immunity in domestic dog populations in the context of control strategies. Vaccination may reduce the predicted influence of naturally acquired immunity on dynamics, as the resulting reduction in transmission will reduce exposure and therefore the expected seroprevalence. As a result of this, and the uncertainty over the implications of unvaccinated seropositives, high seroprevalence within a population should not be taken as evidence that lower vaccination coverage is required to reach herd immunity and eradicate rabies. Findings from this study also bring into question the use of serology for rabies surveillance. While serology could provide a mechanism to conduct rabies surveillance that does not rely on reporting of clinical cases, the challenges of interpreting rabies serology, in addition to the cost of implementation, limits the feasibility of this strategy. However, given the number of studies reporting high rabies seroprevalence, across a wide geographical range [16], the potential for high rates of transmission leading to sub-clinical exposure should be considered, in particular as higher R₀ values may indicate the need for higher vaccination coverage for effective control.

There were a number of limitations to the modelling methods used. In the model, Laikipia was treated as an isolated unit. In reality, the county is continuous with other areas with domestic dog populations. Potentially, persistence of rabies occurs at a wider scale with introductions of rabies from other populations sustaining the disease within Laikipia, as shown in domestic dog populations in N'Djamena, Chad and Bangui, Central African Republic [43,45]. Further data are required to improve parameterisation of human-mediated dog movement, to assess the influence of both importation from outside the county and movement of dogs within the county [4]. In the model, it was also assumed that contact between infectious individuals in different patches was determined by distance. In reality, physical features and human geography will be influential for contact, for example depending on the presence of roads linking properties, or physical barriers such as fencing or rivers separating them [27]. A further limitation was the failure to capture heterogeneity in certain parameters. Rabies transmission is highly heterogeneous with most individuals not transmitting at all, whereas others infect large numbers of other individuals [9]. A negative binomial distribution has been suggested to be the best fit to model these dynamics [18], however the use of the Gillespie

algorithm to model stochasticity precluded fitting specific distributions. Variation in the duration of the latent period has also been shown to influence rabies persistence [4]. Including this heterogeneity may have increased the probability of persistence of rabies.

In conclusion, our results suggest that subclinical immunizing exposure could play a role in the dynamics of rabies in domestic dogs, limiting disease incidence and population decline. However, consideration of other factors, such as spatial structure and human response to rabid dogs, is also required to approximate realistic rabies dynamics. The scenarios for naturally acquired immunity explored for an R_0 of 1.2 produced low predicted seroprevalences relative to those observed in some empirical studies. Higher seroprevalences could be explained by higher rates of immunizing subclinical infection within domestic dog populations, however false positives or spatial variation in seroprevalence may also contribute. If high seroprevalences to effectively control this disease.

Supporting information

S1 Text. Additional details on model parameterisation. (PDF)

S2 Text. Methods and results for sensitivity analysis of the non-spatial model. (PDF)

S1 Table. Model results for non-spatial model across all 25 parameter combinations, presented as heat maps in Fig 2.

(PDF)

S1 Fig. Influence of incidence-dependent human response on model outputs in the absence of naturally acquired immunity (Scenario A). (PDF)

S2 Fig. Comparison of model outputs from immunity scenarios (A, B and C) across different model formulations for R0 = 1.2. (PDF)

S3 Fig. Comparison of model outputs from immunity scenarios (A, B and C) across different model formulations for R0 = 1.5. (PDF)

S1 Code. This file contains: R script to generate the events data frame for human-mediated dog movement, an R script to run model simulations for both the non-spatial and spatial model, a csv. file containing the contact matrix for the spatial model and the shapefile of patches used.

(ZIP)

Acknowledgments

We thank Dr Stefan Widgren for his advice on using the SimInf package for this work.

Author Contributions

Conceptualization: Susannah Gold, Christl A. Donnelly, Rosie Woodroffe, Pierre Nouvellet.

Formal analysis: Susannah Gold.

Investigation: Susannah Gold.

Methodology: Susannah Gold, Pierre Nouvellet.

Supervision: Christl A. Donnelly, Rosie Woodroffe, Pierre Nouvellet.

Visualization: Susannah Gold.

Writing - original draft: Susannah Gold.

Writing – review & editing: Susannah Gold, Christl A. Donnelly, Rosie Woodroffe, Pierre Nouvellet.

References

- Hampson K, Coudeville L, Lembo T, Sambo M, Kieffer A, Attlan M, et al. Estimating the Global Burden of Endemic Canine Rabies. PLoS Negl Trop Dis. 2015; 9: e0003709. <u>https://doi.org/10.1371/journal.pntd.0003709</u> PMID: 25881058
- Rupprecht CE, Barrett J, Briggs D, Cliquet F, Fooks AR, Lumlertdacha B, et al. Can rabies be eradicated? Dev Biol. 2008; 131: 95–121. PMID: <u>18634470</u>
- 3. World Health Organization. WHO expert consultation on rabies: Second report. World Health Organ Tech Rep Ser. 2013; 982.
- Colombi D, Poletto C, Nakouné E, Bourhy H, Colizza V. Long-range movements coupled with heterogeneous incubation period sustain dog rabies at the national scale in Africa. PLoS Negl Trop Dis. 2020; 14: e0008317. https://doi.org/10.1371/journal.pntd.0008317 PMID: 32453756
- Hampson K, Dushoff J, Bingham J, Brückner G, Ali YH, Dobson A. Synchronous cycles of domestic dog rabies in sub-Saharan Africa and the impact of control efforts. Proc Natl Acad Sci. 2007; 104: 7717–7722. https://doi.org/10.1073/pnas.0609122104 PMID: 17452645
- Kitala PM, McDermott JJ, Coleman PG, Dye C. Comparison of vaccination strategies for the control of dog rabies in Machakos District, Kenya. Epidemiol Infect. 2002; 129: 215–222. https://doi.org/10.1017/ s0950268802006957 PMID: 12211590
- Zinsstag J, Dürr S, Penny MA, Mindekem R, Roth F, Gonzalez SM, et al. Transmission dynamics and economics of rabies control in dogs and humans in an African city. Proc Natl Acad Sci. 2009; 106: 14996–15001. https://doi.org/10.1073/pnas.0904740106 PMID: 19706492
- Cleaveland S, Dye C. Maintenance of a microparasite infecting several host species: rabies in the Serengeti. Parasitology. 1995; 111 Suppl: S33–47. <u>https://doi.org/10.1017/s0031182000075806</u> PMID: 8632923
- Hampson K, Dushoff J, Cleaveland S, Haydon DT, Kaare M, Packer C, et al. Transmission Dynamics and Prospects for the Elimination of Canine Rabies. PLOS Biol. 2009; 7: e1000053. <u>https://doi.org/10. 1371/journal.pbio.1000053</u> PMID: 19278295
- Fekadu M, Baer GM. Recovery from clinical rabies of 2 dogs inoculated with a rabies virus strain from Ethiopia. Am J Vet Res. 1980; 41: 1632–1634. PMID: 7224288
- 11. Gnanadurai CW, Zhou M, He W, Leyson CM, Huang C, Salyards G, et al. Presence of Virus Neutralizing Antibodies in Cerebral Spinal Fluid Correlates with Non-Lethal Rabies in Dogs. PLoS Negl Trop Dis. 2013; 7: e2375. https://doi.org/10.1371/journal.pntd.0002375 PMID: 24069466
- Bell JF, Moore GJ. Susceptibility of carnivora to rabies virus administered orally. Am J Epidemiol. 1971; 93: 176–182. https://doi.org/10.1093/oxfordjournals.aje.a121244 PMID: 5104866
- Ramsden RO, Johnston DH. Studies on the oral infectivity of rabies virus in carnivora. J Wildl Dis. 1975; 11: 318–324. https://doi.org/10.7589/0090-3558-11.3.318 PMID: 1097742
- Moore SM, Gilbert A, Vos A, Freuling CM, Ellis C, Kliemt J, et al. Rabies Virus Antibodies from Oral Vaccination as a Correlate of Protection against Lethal Infection in Wildlife. Trop Med Infect Dis. 2017; 2: 31. https://doi.org/10.3390/tropicalmed2030031 PMID: 30270888
- Smith TG, Millien M, Vos A, Fracciterne FA, Crowdis K, Chirodea C, et al. Evaluation of immune responses in dogs to oral rabies vaccine under field conditions. Vaccine. 2019; 37: 4743–4749. <u>https:// doi.org/10.1016/j.vaccine.2017.09.096</u> PMID: 29054727
- Gold S, Donnelly CA, Nouvellet P, Woodroffe R. Rabies virus-neutralising antibodies in healthy, unvaccinated individuals: What do they mean for rabies epidemiology? PLoS Negl Trop Dis. 2020; 14: e0007933. https://doi.org/10.1371/journal.pntd.0007933 PMID: 32053628

- Gilbert AT, Fooks AR, Hayman DTS, Horton DL, Müller T, Plowright R, et al. Deciphering Serology to Understand the Ecology of Infectious Diseases in Wildlife. EcoHealth. 2013; 10: 298–313. <u>https://doi.org/10.1007/s10393-013-0856-0 PMID: 23918033</u>
- Rajeev M, Metcalf CJE, Hampson K. Chapter 20: Modeling canine rabies virus transmission dynamics. In: Fooks A, Jackson A, editors. Rabies: Scientific Basis of the Disease and Its Management. 2020. pp. 655–670.
- Jemberu WT, Molla W, Almaw G, Alemu S. Incidence of Rabies in Humans and Domestic Animals and People's Awareness in North Gondar Zone, Ethiopia. PLoS Negl Trop Dis. 2013; 7: e2216. <u>https://doi.org/10.1371/journal.pntd.0002216 PMID: 23675547</u>
- Cleaveland S, Kaare M, Tiringa P, Mlengeya T, Barrat J. A dog rabies vaccination campaign in rural Africa: impact on the incidence of dog rabies and human dog-bite injuries. Vaccine. 2003; 21: 1965– 1973. https://doi.org/10.1016/s0264-410x(02)00778-8 PMID: 12706685
- Cleaveland S, Barrat J, Barrat M-J, Selve M, Kaare M, Esterhuysen J. A rabies serosurvey of domestic dogs in rural Tanzania: results of a rapid fluorescent focus inhibition test (RFFIT) and a liquid-phase blocking ELISA used in parallel. Epidemiol Infect. 1999; 123: 157–164. <u>https://doi.org/10.1017/</u> s0950268899002563 PMID: 10487652
- Prager KC, Mazet JAK, Dubovi EJ, Frank LG, Munson L, Wagner AP, et al. Rabies Virus and Canine Distemper Virus in Wild and Domestic Carnivores in Northern Kenya: Are Domestic Dogs the Reservoir? EcoHealth. 2012; 9: 483–498. https://doi.org/10.1007/s10393-013-0815-9 PMID: 23459924
- Bahloul C, Taieb D, Kaabi B, Diouani MF, Hadjahmed SB, Chtourou Y, et al. Comparative evaluation of specific ELISA and RFFIT antibody assays in the assessment of dog immunity against rabies. Epidemiol Infect. 2005; 133: 749–757. https://doi.org/10.1017/s095026880500381x PMID: 16050522
- Laurenson K, Esterhuysen J, Stander P, Van Heerden J. Aspects of rabies epidemiology in Tsumkwe District, Namibia. Onderstepoort J Vet Res. 1997; 64: 39–45. PMID: 9204502
- Kayali U, Mindekem R, Yémadji N, Oussiguéré A, Naässengar S, Ndoutamia AG, et al. Incidence of canine rabies in N'Djaména, Chad. Prev Vet Med. 2003; 61: 227–233. https://doi.org/10.1016/j. prevetmed.2003.07.002 PMID: 14554145
- Kitala PM, McDermott JJ, Kyule MN, Gathuma JM. Community-based active surveillance for rabies in Machakos District, Kenya. Prev Vet Med. 2000; 44: 73–85. <u>https://doi.org/10.1016/s0167-5877(99)</u> 00114-2 PMID: 10727745
- Brunker K, Lemey P, Marston DA, Fooks AR, Lugelo A, Ngeleja C, et al. Landscape attributes governing local transmission of an endemic zoonosis: Rabies virus in domestic dogs. Mol Ecol. 2018; 27: 773– 788. https://doi.org/10.1111/mec.14470 PMID: 29274171
- Beyer HL. Epidemiological models of rabies in domestic dogs: dynamics and control. PhD, University of Glasgow. 2010. Available: https://eleanor.lib.gla.ac.uk/record=b2768008
- 29. Blackwood JC, Streicker DG, Altizer S, Rohani P. Resolving the roles of immunity, pathogenesis, and immigration for rabies persistence in vampire bats. Proc Natl Acad Sci. 2013; 110: 20837–20842. https://doi.org/10.1073/pnas.1308817110 PMID: 24297874
- Coleman PG, Dye C. Immunization coverage required to prevent outbreaks of dog rabies. Vaccine. 1996; 14: 185–186. https://doi.org/10.1016/0264-410x(95)00197-9 PMID: 8920697
- Rupprecht CE, Hanlon CA, Hemachudha T. Rabies re-examined. Lancet Infect Dis. 2002; 2: 327–343. https://doi.org/10.1016/s1473-3099(02)00287-6 PMID: 12144896
- **32.** Constantine DG. Rabies transmission by nonbite route. Public Health Rep. 1962; 77: 287–289. PMID: 13880956
- Delpietro H, Segre L, Marchevsky N, Berisso M. Rabies transmission to rodents after ingestion of naturally infected tissues. Medicina (Mex). 1990; 50: 356–360. PMID: 2130231
- Morters MK, Restif O, Hampson K, Cleaveland S, Wood JLN, Conlan AJK. Evidence-based control of canine rabies: a critical review of population density reduction. J Anim Ecol. 2013; 82: 6–14. https://doi. org/10.1111/j.1365-2656.2012.02033.x PMID: 23004351
- Townsend SE, Sumantra IP, Pudjiatmoko, Bagus GN, Brum E, Cleaveland S, et al. Designing Programs for Eliminating Canine Rabies from Islands: Bali, Indonesia as a Case Study. PLoS Negl Trop Dis. 2013; 7: e2372. https://doi.org/10.1371/journal.pntd.0002372 PMID: 23991233
- 36. Cliquet F, Verdier Y, Sagné L, Aubert M, Schereffer JL, Selve M, et al. Neutralising antibody titration in 25,000 sera of dogs and cats vaccinated against rabies in France, in the framework of the new regulations that offer an alternative to quarantine. Rev Sci Tech Int Off Epizoot. 2003; 22: 857–866. <u>https://doi.org/10.20506/rst.22.3.1437</u> PMID: 15005543
- Minke JM, Bouvet J, Cliquet F, Wasniewski M, Guiot AL, Lemaitre L, et al. Comparison of antibody responses after vaccination with two inactivated rabies vaccines. Vet Microbiol. 2009; 133: 283–286. https://doi.org/10.1016/j.vetmic.2008.06.024 PMID: 18757142

- Tepsumethanon W, Polsuwan C, Lumlertdaecha B, Khawplod P, Hemachudha T, Chutivongse S, et al. Immune response to rabies vaccine in Thai dogs: A preliminary report. Vaccine. 1991; 9: 627–630. https://doi.org/10.1016/0264-410x(91)90186-a PMID: 1950096
- Dodds WJ, Larson LJ, Christine KL, Schultz RD. Duration of immunity after rabies vaccination in dogs: The Rabies Challenge Fund research study. Can J Vet Res. 2020; 84: 153–158. PMID: 32255911
- 40. Moore MC, Davis RD, Kang Q, Vahl CI, Wallace RM, Hanlon CA, et al. Comparison of anamnestic responses to rabies vaccination in dogs and cats with current and out-of-date vaccination status. J Am Vet Med Assoc. 2015; 246: 205–211. https://doi.org/10.2460/javma.246.2.205 PMID: 25554936
- Coyne MJ, Burr JHH, Yule TD, Harding MJ, Tresnan DB, McGavin D. Duration of immunity in dogs after vaccination or naturally acquired infection. Vet Rec. 2001; 149: 509–515. <u>https://doi.org/10.1136/vr.</u> 149.17.509 PMID: 11708635
- Woodroffe R, Donnelly CA. Risk of contact between endangered African wild dogs Lycaon pictus and domestic dogs: opportunities for pathogen transmission. J Appl Ecol. 2011; 48: 1345–1354. <u>https://doi.org/10.1111/j.1365-2664.2011.02059.x</u>
- Laager M, Mbilo C, Madaye EA, Naminou A, Léchenne M, Tschopp A, et al. The importance of dog population contact network structures in rabies transmission. PLoS Negl Trop Dis. 2018; 12: e0006680. https://doi.org/10.1371/journal.pntd.0006680 PMID: 30067733
- 44. Ferguson EA, Hampson K, Cleaveland S, Consunji R, Deray R, Friar J, et al. Heterogeneity in the spread and control of infectious disease: consequences for the elimination of canine rabies. Sci Rep. 2015; 5: 18232. https://doi.org/10.1038/srep18232 PMID: 26667267
- Bourhy H, Nakouné E, Hall M, Nouvellet P, Lepelletier A, Talbi C, et al. Revealing the Micro-scale Signature of Endemic Zoonotic Disease Transmission in an African Urban Setting. PLOS Pathog. 2016; 12: e1005525. https://doi.org/10.1371/journal.ppat.1005525 PMID: 27058957
- Center For International Earth Science Information Network. Gridded Population of the World, Version 4 (GPWv4): Population Count, Revision 11. Palisades, NY: NASA Socioeconomic Data and Applications Center (SEDAC); 2018. https://doi.org/10.7927/H4JW8BX5
- 47. Spatial data and maps–Mpala. [cited 9 Mar 2021]. Available: https://mpala.org/data/spatial-data-andmaps/
- Widgren S, Bauer P, Eriksson R, Engblom S. SimInf: An R package for Data-driven Stochastic Disease Spread Simulations. ArXiv160501421 Q-Bio Stat. 2018 [cited 12 Nov 2020]. Available: http://arxiv.org/ abs/1605.01421
- 49. Conan A, Akerele O, Simpson G, Reininghaus B, Rooyen J van, Knobel D. Population Dynamics of Owned, Free-Roaming Dogs: Implications for Rabies Control. PLoS Negl Trop Dis. 2015; 9: e0004177. https://doi.org/10.1371/journal.pntd.0004177 PMID: 26545242
- Morters MK, McKinley TJ, Restif O, Conlan AJK, Cleaveland S, Hampson K, et al. The demography of free-roaming dog populations and applications to disease and population control. J Appl Ecol. 2014; 51: 1096–1106. https://doi.org/10.1111/1365-2664.12279 PMID: 25657481
- Beyer HL, Hampson K, Lembo T, Cleaveland S, Kaare M, Haydon DT. Metapopulation dynamics of rabies and the efficacy of vaccination. Proc R Soc B Biol Sci. 2011; 278: 2182–2190. <u>https://doi.org/10. 1098/rspb.2010.2312</u> PMID: 21159675
- Laager M, Léchenne M, Naissengar K, Mindekem R, Oussiguere A, Zinsstag J, et al. A metapopulation model of dog rabies transmission in N'Djamena, Chad. J Theor Biol. 2019; 462: 408–417. https://doi. org/10.1016/j.jtbi.2018.11.027 PMID: 30500602
- Hayman DTS, Luis AD, Restif O, Baker KS, Fooks AR, Leach C, et al. Maternal antibody and the maintenance of a lyssavirus in populations of seasonally breeding African bats. Rupprecht CE, editor. PLOS ONE. 2018; 13: e0198563. https://doi.org/10.1371/journal.pone.0198563 PMID: 29894488
- Dimitrov DT, Hallam TG, Rupprecht CE, Turmelle AS, McCracken GF. Integrative models of bat rabies immunology, epizootiology and disease demography. J Theor Biol. 2007; 245: 498–509. https://doi.org/ 10.1016/j.jtbi.2006.11.001 PMID: 17184793
- 55. Gilbert A, Greenberg L, Moran D, Alvarez D, Alvarado M, Garcia DL, et al. Antibody response of cattle to vaccination with commercial modified live rabies vaccines in Guatemala. Prev Vet Med. 2015; 118: 36–44. https://doi.org/10.1016/j.prevetmed.2014.10.011 PMID: 25466762
- Moore SM, Hanlon CA. Rabies-Specific Antibodies: Measuring Surrogates of Protection against a Fatal Disease. PLoS Negl Trop Dis. 2010; 4: e595. <u>https://doi.org/10.1371/journal.pntd.0000595</u> PMID: 20231877
- 57. Ferguson AW, Muloi D, Ngatia DK, Kiongo W, Kimuyu DM, Webala PW, et al. Volunteer based approach to dog vaccination campaigns to eliminate human rabies: Lessons from Laikipia County, Kenya. PLoS Negl Trop Dis. 2020; 14: e0008260. https://doi.org/10.1371/journal.pntd.0008260 PMID: 32614827